Author's response to reviews

Title: Hepatitis B vaccinations among Koreans: Results from 2005 Korea National Cancer Screening Survey

Authors:

Hee-Soon Juon (hjuon@jhsph.edu)
Kui Son Choi (kschoi@ncc.re.kr)
Eun-Cheol Park (ecpark@ncc.re.kr)
Min-Son Kwak (serene@ncc.re.kr)
Sunmin Lee (sunmin@umd.edu)

Version: 7 Date: 20 November 2009

Author's response to reviews:

Dear editor.

I found some formatting problems with my manuscript. I made some changes about this formatting problems and typographical errors.

Thanks.

Kui Son Choi Ph.D.

National Cancer Control Institute,
National Cancer Center, Korea
Tel: 82-31-920-2912
Fax: 82-31-920-2189
E-mail: kschoi@ncc.re.kr